Author's response to reviews

Title: Local steroid injection for moderately severe idiopathic carpal tunnel syndrome: Protocol of a randomized double-blind placebo-controlled trial (NCT 00806871)

Authors:

   Magnus Flondell (magnus.flondell@med.lu.se)
   Manfred Hofer (manfred.hofer@skane.se)
   Jonas Bjork (jonas.bjork@skane.se)
   Isam Atroshi (isam.atroshi@med.lu.se)

Version: 5 Date: 14 April 2010

Author's response to reviews:

Dear Editor,

we have made corrections in the authors' affiliations.

Sincerely Magnus Flondell